





ORIGINAL ARTICLE OPEN ACCESS

Synergistic Gains and National Cultural Distance

 $^1University \ of \ Essex, Essex, UK \ \mid \ ^2University \ of \ Vaasa, \ Vaasa, \ Finland \ \mid \ ^3Swansea \ University, Swansea, UK \ \mid \ ^2University \ of \ Vaasa, \ Vaasa, \ Finland \ \mid \ ^3Swansea \ University, Swansea, UK \ \mid \ ^2University \ of \ Vaasa, \ Va$

Correspondence: Tanveer Hussain (th22205@essex.ac.uk)

Received: 21 April 2025 | Revised: 30 July 2025 | Accepted: 6 August 2025

Keywords: cross-border mergers and acquisitions | national cultural distance | organizational learning theory | stock payment | synergistic gains | takeover experience | G34 | D40

ABSTRACT

Does national cultural distance create higher synergistic gains in cross-border mergers and acquisitions (CBMAs)? Existing research on the role of cultural distance suggests that cultural disparities destroy shareholders' wealth. Using an international sample of CBMAs over 19 years, we document that synergistic gains increase by 1.75 percentage points with one standard deviation increase in cultural distance. Drawing from the organizational learning theory, we suggest that learning diverse cultural practices in the post-acquisition stage is a source of higher synergy gains. The positive association between cultural distance and synergies is more pronounced if the acquirer pays in stock and already has takeover experience. This suggests that better awareness of the target country's culture and risk management through stock payment are boundary conditions for higher gains. Overall, our results lead to the counter-intuitive finding that CBMAs between firms from countries with dissimilar cultures are not always valued as destructive but depend on how merging firms learn the cultural practices of one another and manage integration challenges. We offer practical implications for regulators and policymakers about how the international takeover market can serve as a vehicle for learning new cultural practices and increasing combined firm value.

JEL Classification: G34, D40s

1 | Introduction

Takeover synergies are expected around the deal announcements in cross-border mergers and acquisitions (henceforth, CBMAs). From the lens of social identity theory and the cultural adversity hypothesis (Hoffman et al. 1996; Tajfel and Turner 2004), contemporary research on CBMAs delineates that national cultural differences between the acquirer and target home countries negatively affect synergies as an outcome of conflict, distrust, uncertainty, and post-integration challenges (Ahern et al. 2015; Aybar and Ficici 2009; Graebner et al. 2017; Wang et al. 2020). However, the organizational learning theory suggests such dissimilarities may be discerned as tempting (Brock 2005; Very et al. 1996; Reus and Lamont 2009) since they can be a provenance of learning opportunities and value-creation (Stahl and Voigt 2008; Weber and Tarba 2012; Chikhouni et al. 2017). In a broader

context, the asset of foreignness (AOF) concept suggests that apart from the losses from foreign operations, benefits can also be experienced depending on how effectively combining firms tackle post-integration challenges (Lee et al. 2022; Sethi and Judge 2009). Although synergy realization is an important motive behind CBMAs, it has not gained much attention in examining the association between the national culture distance and synergistic gains. Thus, understanding the national cultural disparity² is indispensable for generating synergies and ensuring the long-term success of CBMAs.

Considering the importance of cultural differences in CBMAs, we address an important research question of whether the national culture distance³ affects synergistic gains⁴. This question is pivotal to investigating the sources of higher combined firms' returns, as firms vary significantly in terms of cultural practices

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

© 2025 New York University Salomon Center.

across the border (Arouri et al. 2019). The cultural practices, values, and norms that are rampant in one nation can affect the way organizations are operated, managed, and structured, and the learning of such practices is important in developing effective strategies to adopt diverse cultural practices for promoting ex-post integration in CBMAs. In contrast to existing CBMAs studies, we propose that the cultural distance is not necessarily associated with value destruction but creates an opportunity for sharing and adopting diverse cultural practices in the post-acquisition stage, where two types of cultures are exposed to each other and learning new cultural practices is a source of higher synergy gains.

To test our hypotheses, we used an international sample of CBMAs from 2003 to 2021 to examine the effect of the national cultural distance on synergistic gains. If the learning of cultural practices holds, we expect a positive association between the country-level cultural distance and the synergy gains. Our results show that the national cultural distance positively affects synergistic gains, corroborating organizational learning theory. Economically, a one-standard-deviation increase in the cultural distance⁵ improves takeover synergies by 1.75 percentage points (pp). The individual returns to the acquirer and target shareholders also increase; hence, we propose that takeover value is shared between the merging firms. We further find that the distance-synergy relationship only holds if the payment is made in stock and the acquirer already has takeover experience. It is so because target shareholders become the investors of the combined firm in stock finance deals (Starks and Wei 2013), and the takeover experience of the acquirer enables them to better manage post-deal integration challenges (Chao 2018; Cuypers et al. 2017; Laamanen and Keil 2008). Our results hold after controlling for deal characteristics, country-level characteristics, and combining firm characteristics. We pass a battery of robustness tests, including alternative measures of national culture and synergistic gains, using subsamples, and controlling for sample selection bias.

This study makes three significant contributions to finance literature by examining the role of cultural differences in CBMAs (Kroon et al. 2015; Very et al. 1996; Wang et al. 2020). First, despite the extant work on the cultural adversity hypothesis and social identity theory (Bereskin et al. 2018; Hoffman et al. 1996; Tafjel and Turner 2004), we know little about whether national cultural practices can be learnt through CBMAs. Thus, we support studies on organizational learning theory (Brock 2005; Chikhouni et al. 2017) and the AOF (Lee et al. 2022; Mallon and Fainshmidt 2017) by presenting the evidence that countries differ considerably in cultural values worldwide, and this disparity affects how the returns from CBMAs accrue to shareholders of the combined firm. Therefore, this work offers a better understanding of national cultural dissimilarities for making value-enhancing deals in worldwide takeovers.

Second, we add to studies on how synergy gains are shared between shareholders of acquirers and targets (Martynova and Renneboog 2008; Hussain and Shams 2022; Wang and Xie 2009) by showing the symmetric distribution of returns and arguing that takeover value is shared between the acquirer and target shareholders in CBMAs. Importantly, our study supports the findings of Wang and Xie (2009) and suggests that apart from the difference in firm-level governance, the national cultural distance

also explains the sharing of takeover benefits. This finding suggests that the national cultural distance not only increases synergistic gains but also the way these gains are distributed between the combining firms.

Third, we enhance the understanding of the role of takeover experience and payment method (Aktas et al. 2011; Graham et al. 2002; Jaffe et al. 2013; Renneboog and Vansteenkiste 2019; Starks and Wei 2013). We argue that the learning of diverse cultural practices is more relevant in stock-financed deals where target shareholders become the partners of the combined firm. Similarly, acquirers with takeover experience effectively manage integration problems (Cuypers et al. 2017; Laamanen and Keil 2008). Overall, our findings offer practical implications for regulators and policymakers about how the international takeover market can serve as a vehicle for learning new cultural practices and increasing combined firm value.

The remaining study is arranged as follows: Section 2 presents a literature review and develops hypotheses; Section 3 describes data and shows summary statistics; Section 4 describes methodology; Section 5 discusses results; Section 6 reports additional analysis; Section 7 shows robustness tests; Section 8 concludes the study.

2 | Literature Review and Hypotheses Development

2.1 | Synergy Motive and CBMAs

The synergy motive (Berkovitch and Narayanan 1993) presumes that managers of acquirers and targets maximize shareholder wealth by engaging in takeovers that have the potential for value creation. In simpler terms, synergy means when the value of the combined firm is higher than the individual values of the combining firms (Goold and Campbell 1998). Thus, it follows that returns to both sets of shareholders should be positive if the motive behind a deal is to gain synergies. However, existing studies rarely observe positive synergistic gains around the deal announcement, notable exceptions being Hussain and Shams (2022) and Wang and Xie (2009). These studies argue that higher differences in firm-level governance or corporate social responsibility increase combined returns to acquirers and targets. Most studies show that acquirers, on average, lose shareholder value while target shareholders get benefits mainly because acquirer managers either overpay or show hubristic behaviour (Alexandridis et al. 2010; Andrade et al. 2001; Boone and Mulherin 2007; Renneboog and Vansteenkiste 2019).

The takeover synergies are contingent on the interaction of combining firms' economic fundamentals of increasing revenues or decreasing costs. Since the takeover synergies depend on economic fundamentals, merging firms could favour cross-border deals, as the potential of synergies is higher in such deals despite the existence of higher frictions (Ahern et al. 2015; Martynova and Renneboog 2008). Acknowledging the popularity of CBMAs and their effect on reapportioning economic activity (Col 2017; Erel et al. 2012), it is essential to discern elements that influence a firm's decision to expand its operations across the border. It is highly debated that country characteristics might motivate

firms to enter into other markets through CBMAs. Earlier studies recommend that takeover value in CBMAs can be attributed to geographic proximity (Ahern et al. 2015), different tax systems and governance standards (Col 2017; Ellis et al. 2017; Martynova and Renneboog 2008), and acquirer's currency appreciation (Erel et al. 2012). Also, CBMAs could generate takeover synergies as they offer higher growth potential in foreign markets, have access to better distribution channels, and remove managerial deficiencies (Wang et al. 2020; Ahern et al. 2015), amongst others.

2.2 | AOF

In a broader economic theory, existing CBMA literature posits foreignness as a liability (Taussig 2017; Wu and Salomon 2017) or asset (Lee et al. 2022; Mallon and Fainshmidt 2017). The concept of AOF has been studied using the lens of several theories, including portability theory, resource-based view (RBV), transaction cost theory, and organizational learning theory. To comprehend AOF, each theory underscores the foreign firms' strategies to tackle obstacles associated with the host country's market. The advocates of portability theory and RBV argue that firms can transfer valuable resources from one firm to another through M&As (Cuervo-Cazurra et al. 2007; Hussain et al. 2024; Ellis et al. 2017), whereas others (Ahammad et al. 2017; Boeh 2011) use transaction cost theory and argue that firms operate in other countries considering the cost-benefit analysis of foreign business. The organizational learning theory suggests that pre-deal dissimilarities between the bidder and target provide learning opportunities (Brock 2005; Chikhouni et al. 2017; Reus and Lamont 2009) which can generate synergistic gains.

2.3 | National Cultural Differences and CBMAs Performance

In the past, CBMAs research has emphasized the cultural differences between the acquirers' and targets' home countries and has examined how this gap negatively influences takeover outcomes. A commonly held view is that cultural distance would create information asymmetries (López-Duarte and Vidal-Suárez 2010; Petersen et al. 2008) and complications in the transfer of capabilities, resources, practices, and skills (Björkman et al. 2007; Park and Ungson 1997). Due to a poor understanding of the target's country culture—social beliefs, assumptions, values, and norms—firms have to bear higher costs for acquiring and learning information, undertaking cross-cultural communications, and adopting a diligence process (Konara and Mohr 2023) to comply with the new cultural environment. This stream of literature uses theories of self-categorization and social identity to support the negative association between cultural disparities and takeover performance and argues that such differences result in conflict, distrust, limited collaboration, and resistance.

Despite most of the CBMAs literature finding the negative consequences of cultural disparity, contemporary research has started to highlight the positive impacts of cultural distance (Björkman et al. 2007; Stahl et al. 2016; Stahl and Tung 2015). This strand of literature contends that cultural distance does not necessarily lead to poor takeover performance but creates room for knowledge sharing and learning. Distance in culture

dispenses a foundation for differentiation, and the higher the cultural distance between the home countries of the acquirers and targets, the larger the gaps in terms of repertoires and routines (Morosini et al. 1998), management styles, strategic orientations, organization structures, business practices, and firm values (Dhanaraj and Beamish 2004; Hitt et al. 1997; Lubatkin et al. 1998). These studies use a cross-cultural learning perspective to highlight the positive effect of cultural distance and takeover outcomes and recommend that CBMAs may allow combining firms to get additional benefits due to differences in their national cultural practices.

2.4 | Hypotheses Development

National culture plays an important role in shaping the practices of communities and refers to a wide range of traditions, beliefs, and customs, amongst others, that are distinctive to a particular society (Wang et al. 2020; Ahern et al. 2015). Yet, in the contemporary globalized world, growth through CBMAs permits firms to benefit from one another to realize takeover synergies (Ellis et al. 2017; Martynova and Renneboog 2008). An important source of synergistic gains in CBMAs may be induced by the learning of diverse cultural practices in the combined firm. Drawing from organizational learning theory (see, among others, Curado 2006; Drejer 2000), we propose that cultural values can be shared and learnt in the combined firm and that a higher distance between the national culture of combining firms is associated with higher synergistic gains. Also, we assume that firms from countries with different cultures have the potential to learn cultural practices so that the combined firm does not face intense integration challenges. As diverse culture is anticipated to create additional value, the synergistic gains must reflect such value creation.

The theoretical economic logic suggesting that cultural distance can increase synergistic gains revolves around the role of diverse routines, repertoires, norms, and other managerial aspects (Morosini et al. 1998; Reus and Lamont 2009) in CBMAs. Although, an important strand of literature on cultural distance (Aybar and Ficici 2009; Graebner et al. 2017; Wang et al. 2020) shows that the differences in certain routines destroy postacquisition performance, another body of literature (Brock 2005; Chikhouni et al. 2017; Slangen 2006) suggests that cultural distance is not always harmful to the performance of multinational firms. In a broader context, repertoires and routines are ways of managing business activities, and integrating or learning diverse routines may result in improved problem-solving, innovative abilities, and a better understanding of the host country's market dynamics and consumer preferences. Also, multinationals need to hold diverse repertoires and routines to compete in the multicultural world (Morosini et al. 1998), and thus learning new cultural practices can ensure the long-term growth of the new entity. Aftermath benefits of takeovers are contingent on the combining firms' ability to manage cultural clashes between them, whereby better management of clashes can improve postacquisition performance (i.e., synergistic gains).

Slangen and Hennart (2008) argue that takeover benefits can be enjoyed if the cultural distance between the acquirer and target is higher, underscoring the importance of host market knowledge. Several others suggest that established routines are portable

across different countries (Björkman et al. 2007; Ellis et al. 2017; Gupta and Govindarajan 2000) and that satisfying customers with different cultural norms is value-enhancing (Shi et al. 2017; Tan and Mahoney 2003). Thus, cross-border M&As provide a platform to learn new practices and access valuable resources immersed in other cultures without developing gradually over time (Jemison and Sitkin 1986; Kroon et al. 2015; Wang et al. 2020). Based on the learning argument, we present our first hypothesis as follows:

Hypothesis 1. (H1): National cultural distance is positively associated with synergistic gains, ceteris paribus.

M&A literature shows that one of the critical strategic choices is the selection of stock payment, as it incentivizes the target through a risk-sharing mechanism not to hide any bad news that may affect post-deal performance (Boone et al. 2014; Dang et al. 2022; Kanungo 2021). This is especially so in cross-border deals where the levels of information asymmetries are higher (Boeh 2011; Reddy and Fabian 2020); several scholars suggest that the selection of payment method (stock, cash, or a combination of both) depends on the level of information asymmetry (Cuypers et al. 2017; Luypaert and Caneghem 2017), with the acquirer showing a preference for stock where the target is from another country to manage the risk of asymmetric information. If the acquirer cannot collect accurate information about the target, the former will overpay due to an inaccurate valuation of the target's resources (Chari and Chang 2009; Dang et al. 2022), which may result in higher transaction costs and will lead to poor takeover performance.

Notably, Mukherji et al. (2013) suggest that asymmetric information is a key source of overpayment that destroys acquirer wealth. Specifically, the overpayment risk of the acquirer is mitigated with stock financing, as the target would share the risk post-acquisition, and therefore, stock payment is a tool for risk-sharing in the presence of higher information asymmetry (Alexandridis et al. 2012; Shleifer and Vishny 2003). As noted above, shareholders in a target firm may have little interest in promoting the sharing of bad news on a target firm if they hope to get rid of a lemon. However, their grasp of information on the acquiring firm may be limited. Hence, they may encourage mutual information sharing to avoid being saddled with an even riskier consolidated firm. In any event, the risks of overpayment will be shared between both parties (Huang et al. 2016).

Moreover, the higher risk of non-completion in M&As paid with stock would suggest that such deals will only go through if each side is satisfied with the information at their disposal (Huang et al. 2016). Accordingly, we propose that the combined effect of cultural distance and stock payment on synergistic gains should be higher, as the stock payments enable acquirers to share the foreignness risk with targets, avoid overpayment, and better align the interests of the acquirer and the target. Based on this, we propose the following hypothesis:

Hypothesis 2. (H2): Positive association between national cultural distance and synergistic gains should be higher for stock-financed deals, ceteris paribus.

A growing body of literature on takeover experience (proxied by serial acquirers) states that experienced acquirers can derive higher takeover (Chao 2018; Cuypers et al. 2017; Laamanen and Keil 2008). These studies suggest that experience improves acquirers integrating capabilities with every successful takeover, and they do not face the same integration challenges as non-experienced acquirers. Also, such acquirers are well-acquainted with other takeover processes, including target selection and deal negotiation (Aktas et al. 2011; Jaffe et al. 2013; Renneboog and Vansteenkiste 2019). Being repeatedly engaged in takeovers, experienced acquirers have more potential for value creation, and therefore, we can expect that the cultural distance–synergy relationship should be more pronounced if the acquirer has takeover experience. This discussion leads to our second hypothesis:

Hypothesis 3. (H3): Positive association between national cultural distance and synergistic gains should be higher for acquirers having takeover experience, ceteris paribus.

3 | Data and Summary Statistics

The M&A data come from Refinitiv (Thomson Reuters) Securities Data Corporation (SDC) and cover 19 years from 2003 to 2021. We take completed CBMAs deals where acquirers and targets are publicly traded firms whose daily stock price data is available from the DataStream (Hussain and Kumar 2025). The countries with fewer than five deals are eliminated to avoid noise in the analysis. We further exclude deals where acquirers or targets are from the financial and utilities industries, as these industries have different regulations. We ended up with 1917 CBMAs with available data on all variables used in the regression analyses.

Table 1 reports the distributions by the year of deal announcement, acquirer's industry (classified using Fama–French 48-industry), and acquirer's country including the number of deals and percentage of deals. Panel A of Table 1 shows that the most active years in the CBMAS are 2014 (168 deals), 2015 (166 deals), and 2008 and 2011 (both having 161 deals). Panel B shows that 16.95% of acquirers are from the business services industry, 8.09% from electronic equipment, and 6.05% from the chemical industry. Panel C documents that the United State, France, and Germany are leading acquirer countries in our sample of CBMAs with 772, 290, and 243 deals, respectively. We also present sample distribution by target country in Appendix B, showing that 21.80% of target firms in our sample are from the United States, followed by the United Kingdom (11.48%) and Canada (7.77%).

Table 2 shows summary statistics of all variables used in this paper. The average synergistic gains are 4% with a substantial standard deviation of 14%, as shown in Panel A of Table 2. Panel B documents that the average national cultural score difference between firms from different countries is 15.52. Panel C manifests deal-related characteristics, and we find that mostly acquirers pay in stock (15%) and engage in different industries (65%). The average values of the acquirers' (targets') leverage, liquidity, and Tobin's Q are 0.18 (0.02), 0.09 (0.80), and 0.99 (0.04), respectively, as reported in Panel D. We define all involved variables in Appendix A. We also find a positive and statistically significant correlation between synergistic gains and national

TABLE 1 | Sample distributions.

Year	N	%	Acquirer's industry (top 10)	Z	%	Acquirer country	N	%
2003	38	1.98	Business services	325	16.95	Austria	13	0.68
2004	59	3.08	Electronic equipment	155	8.09	Belgium	18	0.94
2005	95	4.96	Chemicals	116	6.05	Brazil	15	0.78
2006	133	6.94	Communication	94	4.90	Canada	101	5.27
2007	143	7.46	Machinery	94	4.90	Denmark	21	1.10
2008	161	8.40	Measuring and control equipment	93	4.85	Finland	61	3.18
2009	103	5.37	Medical equipment	92	4.80	France	290	15.13
2010	135	7.04	Business supplies	83	4.33	Germany	243	12.68
2011	161	8.40	Automobiles and trucks	29	3.50	Italy	18	0.94
2012	134	66.9	Computers	59	3.08	Mexico	10	0.52
2013	109	5.69				Norway	50	2.61
2014	168	8.76				Spain	17	0.89
2015	166	8.66				Sweden	33	1.72
2016	149	7.77				Switzerland	153	7.98
2017	45	2.35				United Kingdom	102	5.32
2018	36	1.88				United States	772	40.27
2019	48	2.50						
2021	34	1.77						
Total	1917	100		1178	61.45		1917	100

Note: This table shows sample distributions by year of the deal announcement, acquirer's industry, and country, including the total number of announced deals (N) and percentage of deals (%) as a proportion of the total announced deals. Our sample consists of CBMAs of listed acquirers and targets reported in Securities Data Corporation (SDC) over the period of 19 years (2003–2021). We consider completed deals, utilize the Fama-French 48 industries, and exclude deals where the acquirer or target firms are from the financial sector (SIC codes 6000-6999) and the utilities sector (SIC codes 4900-4949). 1468016, 0, Downloaded from https://onlinelibrary.wiley.com/doi/10.1111//mii.70002 by INASPHINAR1 - CAMEROON, Wiley Online Library on [2708/2025]. See the Terms and Conditions (https://onlinelibrary.wiley.com/rerms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

TABLE 2 | Descriptive statistics.

	N	Mean	Median	SD	р5	p95
Panel A: Takeover gains						
Synergistic gains	1917	0.04	0.01	0.14	-0.07	0.17
Panel B: Pre-deal differences						
National cultural distance	1917	15.52	7.08	17.56	0.37	49.53
Firm culture gap	1917	58.43	65.00	18.1	17.6	79.19
National governance gap	1917	59.18	60.00	14.94	28.6	87.18
Panel C: Deal characteristics						
Related industry dummy	1917	0.35	0.00	0.48	0.00	1.00
Payment method dummy	1917	0.15	0.00	0.36	0.00	1.00
Number of competing acquirers	1917	1.01	1.00	0.11	1.00	1.00
Panel D: Acquirers and target characteristics						
Run-up	1917	0.01	0.00	0.13	-0.16	0.19
Acquirer leverage	1917	0.18	0.16	0.12	0.01	0.39
Target leverage	1917	0.02	0.00	0.08	0.00	0.17
Acquirer liquidity	1917	0.09	0.07	0.09	0.01	0.26
Target liquidity	1917	0.80	0.80	0.02	0.77	0.83
Acquirer Tobin Q	1917	0.99	0.81	0.55	0.48	1.91
Target Tobin Q	1917	0.03	0.02	0.07	0.01	0.04
Panel E: Acquirer country characteristics						
GDP growth	1917	1.68	1.95	1.88	-2.54	3.93
GDP per capita	1917	10.75	10.76	0.30	10.38	11.33

Note: The table presents summary statistics for the variables involved in the study. Our sample covers all reported in Securities Data Corporation (SDC) over 19 years, where combining firms are publicly listed. The synergistic gains are calculated using the value-weighted portfolio of the acquirer and target gains. The gains are 3-day cumulative abnormal returns around the deal announcement, computed using the market model over 255 to 25 days before the deal announcement. The national cultural distance is the difference between the acquirers' and targets' national cultural scores provided by Hofstede (2011). All variables are defined in Appendix A.

cultural distance and report the Pearson correlation matrix in Appendix ${\bf C}$.

4 | Methodology

4.1 | Cumulative Abnormal Returns

The takeover value should be reflected in synergy gains if acquiring a public target is unanticipated. We hypothesize that national cultural disparity should positively affect synergy gains and estimate expected returns using the following market model:

$$R_{bcm} = \alpha_{bc} + \beta_{bc}R_{cm} + \varepsilon_{bcm}, \quad m = -255, ..., -25$$
 (1)

where R_{bcm} is the acquirer or target daily return m from DataStream in country c; R_{cm} shows the daily market index return m for country c; ε_{bcm} denotes the excess return. Following Fama et al.'s (1969) event study methodology, we compute cumulative abnormal returns for a 3-day period (t-1, t+1) around the deal announcement using an estimation window of 255 to 25 days before the deal announcement. The difference between expected and actual daily returns is the cumulative abnormal return. To examine the effect of cultural difference on synergistic gains, the

following cross-sectional regression model is used:

$$Synergistic \ gain \ (-1,+1)_{d,t} = +\beta_1 Culture \ distance_{d,t-1}$$

$$+ \sum \beta_2 Deal \ controls_{d,t}$$

$$+ \sum \beta_3 Firm \ controls_{b,t-1}$$

$$+ \sum \beta_4 Country \ controls_{b,t-1}$$

$$+ \lambda_y + \eta_i + \gamma_n + \varepsilon_{i,t}$$
 (2)

where $Synergistic\ gain\ (-1,+1)_{d,t}$ is the value-weighted portfolio of the acquirer and target returns over 3-day^7 event window where weights are allocated based on the equity's market value 6 days before the deal d; α shows the intercept; $Cultural\ distance\ _{d,t-1}$ is the pre-deal difference between the acquirer's country culture index and target's country culture index for deal d. $Deal\ controls_{d,t}$ shows a vector of deal-related characteristics for deal d at time t, and it includes: payment method, a dummy variable with a value of one if the deal is purely paid in cash and zero otherwise; number of acquirers, the logarithm of the number of competing acquirers in a deal; same industry deal, a dummy variable having a value of one

if the acquirer and target are from the same Fama–French 48 industry and zero otherwise; relative deal size, value of the deal scaled by the acquirer's market value of equity. Firm $controls_{b,t-1}$ denotes a vector of firm-related characteristics for the acquirer and target a year before the deal and includes leverage, Tobin's Q, size, and stock price run-up. $Country\ controls_{b,t-1}$ indicates a vector of country-related characteristics (i.e., country governance gap, GDP growth, and GDP per capita) 1 year before the deal announcement. To control for omitted variables that may affect synergistic gains, we used year, λ_y , industry, η_i , and country, γ_n fixed effects. Finally, to mitigate the impact of outliers, we winsorize synergistic gains and accounting variables by 1% of their distribution tails.

4.2 | National Cultural Distance

The key variable of interest of the study is the national cultural distance between the home countries of acquirers and targets. To calculate the cultural distance, we followed existing studies (Chand et al. 2021; Kashefi-Pour et al. 2020) and used scores (1 to 100) on Hofstede's (2011) six cultural dimensions described below:

- Power Distance Index (PDI): Measures the extent to which people accept that power among individuals, organizations, and institutions is unequally distributed (Campagnolo and Vincenti 2022; Hofstede 1980; Morosini et al. 1998).
- Individualism Versus Collectivism (IVC): Refers to the degree of preference where people only take care of themselves and their immediate family (Hofstede 1980, 2011).
- Masculinity Versus Femininity (MVF): Measures the degree to which a society prioritizes masculinity-related qualities over femininity-related qualities (Ahmad et al. 2022; Hofstede 1980).
- iv. Uncertainty Avoidance (UAI): Measures the degree to which people in a national culture feel threatened by ambiguity and uncertainty in their life and take measures to avoid them (Hofstede 1980, 2011).
- v. Indulgence Versus Restraint (IVR): Measures the degree to which a national culture allows its people to fulfil their human needs and desires versus the degree to which people in a culture control their natural desires and impulses (Hofstede 2011; Minkov and Hofstede 2011).
- vi. Long-Term Orientation Versus Short-Term (LVS): Measures the degree to which people of a country focus on long-term orientation and connect the past with future actions (Chand et al. 2021; Hofstede 2001).

To measure the national culture distance between the home countries of acquirers and targets, we follow the Kogut and Singh (1988) index that is widely used in the international business literature. We calculate this index using Hofstede's six cultural dimensions: masculinity, power distance, individualism,

uncertainty avoidance, long-term, and indulgence:

$$CG_{bt} = \sqrt{\sum_{a=1}^{6} \frac{(B_{ab} - T_{at})^2}{V_a}}$$

where CG_{bt} is the cultural distance between the acquirers country b and target country t. B_{ab} and T_{at} are the values of national cultural dimension a (a=1 to 6) for acquirer country b and target country t, respectively, and V_a shows the variance of each cultural dimension.

4.3 | Control Variables

We control for three groups of variables associated with synergistic gains: deal characteristics, acquirer/target firm characteristics, and country characteristics.

Based on the M&A literature, we first consider deal-level characteristics, including method of payment, number of acquirers, and same-industry deal. The literature shows that the method of payment affects returns—cash-financed deals increase returns, while stock-financed deals reduce returns due to the adverse selection problem (Graham et al. 2002; Hussain and Loureiro 2023; Myers and Majluf 1984). Diversified deals are value-destructive and face integration problems (Morck et al. 1990), whereas same-industry deals generate positive returns because of economies of scale (Masulis et al. 2007). The returns decrease with an increase in the number of competitors (Bradley et al. 1988; Schwert 1996; Alexandridis et al. 2010) as acquirers pay hefty premiums to win the bid auction.

Second, we control for some acquirer and target characteristics such as leverage, Tobin's *Q* (Lang et al. 1991), liquidity (Cornett et al. 2011), and stock price run-up (Golubov et al. 2012; Brigida et al. 2014). The underlying reasons for the inclusion of these firm characteristics are as follows: Managerial discretion decreases with a higher level of leverage (Lang et al. 1991), motivates managers to increase firm performance (Gilson 1990; Tunyi et al. 2025), and positively affects returns (Wang and Xie 2009). The overvalued firms proxied by higher Tobin's *Q* may buy less overvalued targets using their stock (Danbolt et al. 2016). Managers of firms with higher liquidity over-invest unproductively and generate lower returns (Chatterjee et al. 2021; Hussain et al. 2022). The stock price reaction before the deal announcement is negatively associated with the stock price reaction post-acquisition (Golubov et al. 2012).

Finally, we add country characteristics as controls, such as country governance gap, GDP growth, and GDP per capita. We control for the country governance gap (Ellis et al. 2017; Hussain and Tunyi 2025) between the acquirer and target home countries using World Governance Indicators (WGI), which can positively affect returns. The macro-economic variables—GDP growth and log of GDP per capita—show the level of a country's economic development and can affect returns to firm shareholders (Fauver et al. 2017).

5 | Results

5.1 | Synergistic Gains and Cultural Distance

To test our first hypothesis (H1), we estimate cross-sectional regressions of synergistic gains on the national cultural distance along with a set of control variables. If the national cultural distance creates room for learning diverse cultural values, the synergistic gains should be higher, ceteris paribus. We estimate Equation (2) and report results in Table 3. In Model (1) of Table 3, we only include national culture distance as an explanatory variable to make sure that the association between cultural distance and takeover synergies is not derived by the presence of a control variable. We find that cultural distance positively affects synergy at the 1% level, showing that with the increase in cultural distance, combined returns increase, and we attribute this increase to the learning of new cultural practices. Economically, a one standard deviation increase in cultural distance increases synergistic gains by 1.75 pp. In Model (2), we examine the impact of the cultural distance on synergy gains after controlling for deal characteristics, firm characteristics, and country characteristics. The estimated coefficient on the national culture distance is positive and statistically significant. From Models (3) to (8), we used individual dimensions of Hofstede's culture to examine whether differences in some dimensions are more important than others. We find that distances in all dimensions are significantly important for value creation; however, economic magnitudes vary a bit. The results support our first hypothesis (H1) and suggest that a higher national cultural distance is associated with higher synergistic gains, and this association appears due to the learning of cultural values after the acquisition.

Among the controls, we realize that the corporate governance gap positively affects synergy gains, supporting previous studies (Ellis et al. 2017; Martynova and Renneboog 2008; Wang and Xie 2009). Acquirers' payment in stock increases synergistic gains, and it promotes the role of payment method in M&As (among others, see Graham et al. 2002). Other control variables show qualitatively similar effects on gains to what others find (see, among others, Wang and Xie 2009; Starks and Wei 2013).

Our results on national cultural distance demonstrate that differences in national cultural norms across countries are relevant to illustrate the takeover synergies and that cultural values are learnt in CBMAs. We provide counter-intuitive findings to existing studies on CBMAs (for instance, Kroon et al. 2015; Krug and Nigh 2001; Very et al. 1996) and suggest that cultural disparity is not always a hurdle for CBMAs to derive takeover value but creates room for sharing and learning new cultural practices between combining firms.

Additionally, we test whether the national cultural distance affects acquirer or target returns. We re-estimate Equation (2) by replacing synergistic gains with acquirer or target returns, in separate regressions. We find that cultural distance positively affects both acquirer and target returns (Table 4), implying that synergistic gains emerging from learning of national cultural values are equally shared between merging firms. These results provide further insights into the role of national cultural distance, showing that, apart from synergistic gains, individual gains to

merging firms' shareholders are also explained by the cultural distance.

5.2 | Boundary Conditions: Stock-Financed Deals and Takeover Experience

Under our second and third hypotheses, we test whether the stock-financed deal and takeover experience moderate the association between the national cultural distance and synergistic gains. The underlying reason is that takeover experience allows one to tackle integration challenges in a better way (Cuypers et al. 2017; Laamanen and Keil 2008), and the target firm shareholders become the owners of the newly merged firm if the deal payment is made in stock. Therefore, we estimate the following cross-sectional regression model:

$$\begin{aligned} Synergy(-1,+1)_{d,t} &= & \alpha + \beta_1 Cultural \ distance_{d,t-1} + \beta_2 Stock \ payment_{d,t} \\ &+ & \beta_3 Cultural \ distance_{d,t-1} \ Stock \ payment_{d,t} \\ &+ & \sum \beta_4 Deal \ controls_{d,t} \ + \sum \beta_5 Firm \ controls_{b,t-1} \\ &+ & \sum \beta_6 Country \ controls_{b,t-1} \ + \lambda_y \\ &+ & \eta_i + \gamma_n + \varepsilon_{i,t} \end{aligned}$$

where $Stock\ payment\ _{d,t}$ is a dummy variable that equals one if the acquirer pays in stock and zero otherwise. The variable of interest here is the interaction term between the national culture distance and stock payment [Cultural distance __d,t_-1] \times $Stock\ payment\ _{d,t}$] that should show a positive and significant coefficient if stock-financing is beneficial. We again estimate Equation (3) by replacing $Stock\ payment\ _{d,t}$ with $Takeover\ experience\ _{d,t-1}$ with the same set of controls. We further dichotomize our sample into subsamples of CBMAs: stock-financed versus cash or mixed financed and experienced acquirers versus non-experienced acquirers. We expect that results should be more pronounced in the subsample of stock-financed deals and experienced acquirers.

In Models (1) and (2) of Table 5, the estimated coefficients on the interaction terms are significantly positive, confirming our hypotheses (**H2** and **H3**). We find that the average effect of the cultural distance on synergies is higher by 0.13 pp for stock-financed deals (Model 1) and by 0.10 pp for experienced acquirers (Model 2). Also, the results are more pronounced for subsamples of stock-financed deals and experienced acquirers.

Our findings are aligned with the existing work highlighting positive consequences of stock-financed deals (Rhodes-Kropf and Viswanathan 2004; Starks and Wei 2013), especially in CBMAs where information asymmetries are very high and targets conceal the true value of their firms. However, our findings do not align with or support studies arguing that foreign targets are reluctant to accept acquirer stock as payment (Gaughan 2002; Moeller and Schlingemann 2005). We also support studies emphasizing the benefits of takeover experience and suggesting that acquirers learn from deal to deal and perform better in the latter deals (Chao 2018; Cuypers et al. 2017; Laamanen and Keil 2008) rather than those suggesting losses emerging from serially engaging in M&As

TABLE 3 | Synergistic gains and national cultural distance.

Dependent variable:	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Synergistic gains (-1, +1)	Kogut and Singh (1988) index	Kogut and Singh (1988) index	PDI	IVC	MVF	UAI	IVR	LVS
National cultural distance	0.0010***	0.0009***	0.0004***	0.0005***	0.0008***	0.0007***	0.0013***	0.0009***
distance	(3.172)	(3.299)	(2.637)	(2.722)	(3.533)	(3.908)	(3.908)	(3.943)
Firm culture gap	(51272)	-0.0000	-0.0000	-0.0000	-0.0001	0.0000	0.0000	0.0000
8.1		(-0.194)	(-0.050)	(-0.158)	(-0.332)	(0.057)	(0.057)	(0.122)
National governance gap		0.0005*	0.0005*	0.0005*	0.0005*	0.0005*	0.0005*	0.0005*
		(1.688)	(1.812)	(1.700)	(1.838)	(1.665)	(1.665)	(1.869)
Related industry dummy		-0.0202***	-0.0186***			-0.0179***		-0.0175**
-		(-2.942)	(-2.736)	(-3.021)	(-2.905)	(-2.649)	(-2.649)	(-2.562)
Payment method dummy		0.0252**	0.0270**	0.0271**	0.0255**	0.0257**	0.0257**	0.0286**
		(2.080)	(2.178)	(2.199)	(2.101)	(2.080)	(2.080)	(2.249)
Number of competing acquirers		-0.0248	-0.0238	-0.0269	-0.0245	-0.0226	-0.0226	-0.0330**
•		(-1.515)	(-1.493)	(-1.639)	(-1.534)	(-1.396)	(-1.396)	(-2.013)
Run-up		-0.0747***	-0.0718***	-0.0732***	-0.0762***	-0.0711***		-0.0637**
		(-2.901)	(-2.803)	(-2.858)	(-2.966)	(-2.774)	(-2.774)	(-2.495)
Acquirer leverage		0.0428	0.0384	0.0402	0.0395	0.0454*	0.0454*	0.0367
		(1.568)	(1.405)	(1.473)	(1.449)	(1.651)	(1.651)	(1.344)
Target leverage		0.1114***	0.1157***	0.1085***	0.1145***	0.1148***	0.1148***	0.1090***
		(2.880)	(3.026)	(2.862)	(2.969)	(2.997)	(2.997)	(2.838)
Acquirer liquidity		-0.0303	-0.0314	-0.0294	-0.0326	-0.0324	-0.0324	-0.0221
		(-0.720)	(-0.743)	(-0.703)	(-0.763)	(-0.766)	(-0.766)	(-0.515)
Target liquidity		-2.4064**	-2.4379**	-2.3963**	-2.2995**	-2.3387**	-2.3387**	-2.3080**
		(-2.131)	(-2.163)	(-2.122)	(-2.042)	(-2.069)	(-2.069)	(-2.066)
Acquirer Tobin Q		0.0079	0.0083	0.0076	0.0067	0.0082	0.0082	0.0068
		(0.999)	(1.065)	(0.971)	(0.863)	(1.052)	(1.052)	(0.877)
Target Tobin Q		-0.0790	-0.0814	-0.0797	-0.0731	-0.0793	-0.0793	-0.0684
		(-1.049)	(-1.070)	(-1.051)	(-0.980)	(-1.050)	(-1.050)	(-0.898)
GDP growth		0.0022	0.0025	0.0023	0.0025	0.0024	0.0024	0.0020
		(0.705)	(0.791)	(0.737)	(0.784)	(0.766)	(0.766)	(0.625)
GDP per capita		-0.1076**	-0.1111***	-0.1087 ***	-0.1016**	-0.1053**	-0.1053**	-0.1177***
		(-2.549)	(-2.620)	(-2.582)	(-2.414)	(-2.490)	(-2.490)	(-2.771)
Constant	0.0249	3.0046***	3.0469***	3.0100***	2.8390***	2.9135***	2.9135***	3.0300***
	(0.673)	(2.850)	(2.897)	(2.863)	(2.703)	(2.764)	(2.764)	(2.924)
Year, industry, and country FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

(Continues)

TABLE 3 | (Continued)

Dependent variable:	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Synergistic gains (-1, +1)	Kogut and Singh (1988) index	Kogut and Singh (1988) index	PDI	IVC	MVF	UAI	IVR	LVS
N	1917	1917	1917	1917	1917	1917	1917	1917
R^2	0.0947	0.1161	0.1076	0.1117	0.1170	0.1146	0.1146	0.1140

Note: The sample comprises 1917 completed CBMAs reported in Securities Data Corporation (SDC) between 2003 and 2021. The combining firms are publicly listed, and we consider only completed deals. The synergistic gain is the value-weighted portfolio of the acquirer and target gains over a 3-day event window. Our key variable of interest ('National cultural distance') is calculated using Kogut and Singh's (1988) index, which captures the difference between cultural scores of merging firms' home countries. Models (1) and (2) show results for the effect of the national cultural distance and synergistic gains, while Models (3)–(8) show the effect of all individual culture dimensions on synergistic gains. All variables are defined in Appendix A; standard errors are corrected for heteroscedasticity (White 1980), and *t*-statistics are reported in parentheses. We control for year, industry, and country fixed effects in all models, whose coefficients are not reported for brevity.

*, **, and *** show statistical significance levels at 10%, 5%, and 1%, respectively.

or diversifying operations in already targeted foreign markets (Billett and Qian 2008; Fuller et al. 2002).

5.3 | Addressing Endogeneity

Our findings may suffer from the sample selection bias, as firms may only engage in CBMAs when the potential of takeover synergies is higher. To tackle this issue, we generate two comparable samples of deals with high (treatment group) and low national cultural distance (control group) employing the propensity score matching (PSM). Using the sample median as a cutoff, we split our sample into two groups of high and low cultural distance and apply one-to-one matching with 0.01 calliper distance to identify pairs of comparable CBMAs.

Several studies (Jiang et al. 2018; Shahab et al. 2024) recommend that entropy-balanced matching (EBM) consider an 'equal percent bias reducing' methodology and assign weights based on a repetitive process in a way that the descriptive statistics of treated and control groups are well-aligned (Hainmueller 2013). Specifically, we use the maximum-entropy re-weighting scheme and divide our M&A sample into treated and control) groups. Also, we matched control and treated groups based on the first moment of all covariates used in Table 3.

We acknowledge that engaging in a cross-border M&A deal can be an endogenous choice the acquirer makes, which may affect our inferences. For instance, Rossi and Volpin (2004) show that acquirers from countries with better shareholder protection rights are likely to prefer domestic deals over cross-border deals. Thus, a cross-border takeover will appear only if acquisition gains are high enough to cover additional costs associated with foreign business. This may lead to a positive association between cultural distance and synergistic gains. Accordingly, we use Heckman's (1979) procedure to help control for endogeneity. First, we used the sample of all domestic and cross-border takeovers announced during the sample period and estimated the probability that an acquirer will engage in the cross-border deal rather than the domestic one. Second, we calculate Heckman's λ (Mill's ratio) using the Probit model's resulting parameters for each sample

acquirer. Eventually, we add Heckman's λ as an additional control in our baseline model. After addressing the endogeneity-related issues, our results in Table 6 are consistent with what we reported in baseline models, suggesting that cultural distance increases synergistic gains.

6 | Additional Analyses—Similarities in National Culture

So far, we only provide empirical evidence on the benefits of cultural distance, and in this section, we examine whether the similarity in Hofstede's cultural scores across CBMAs has the predictive power for synergistic gains. We expect that if similar national cultures of merging firms reduce integration issues, then we should observe a positive relationship between cultural similarity and takeover gains. To calculate national cultural similarity, we use Jaffe's (1986) measure for merging firms:

Cultural similarity_{i,j,t} =
$$\frac{X_{i,t}X'_{i,j}}{(X_{i,t}X'_{i,t})^{0.5}(X_{j,t}X'_{i,t})^{0.5'}}$$

where $X_{i,t}$ is a vector of six individual cultural dimensions proposed by Hofstede for acquirer i and $X'_{i,j}$ identifies a vector of the same dimensions for target j. We investigate if there is any impact of cultural similarity on takeover gains, and therefore, we estimate Equation (2) using cultural similarity instead of cultural differences as a key variable of interest. Based on the findings shown in Table 7, high similarity negatively affects synergistic, acquirer, and target gains. These results suggest that similarity destroys takeover value and the distribution of this value between shareholders of merging firms. Importantly, these findings further support our learning argument and highlight that room for learning new cultural values only exists when firms come from countries with different cultures. Furthermore, our results contradict studies on firm-level cultural similarities that explain 'cultural fit' to illustrate better takeover outcomes and lower integration challenges (Ahammad and Glaister 2013; Bereskin et al. 2018; Doukas and Zhang 2021), arguing that national cultural similarities may not matter more in CBMAs.

p < 0.1, p < 0.05, p < 0.01.

TABLE 4 | Acquirer and target gains.

Panel A: Acquirer gains							
Dependent variable:	(1)	(2)	(3)	(4)	(5)	(9)	(2)
Acquirer gains $(-1, +1)$	Kogut and Singh (1988) index	PDI	IVC	MVF	UAI	IVR	IVS
National cultural distance	0.0006***	0.0002**	0.0002**	0.0004***	0.0004***	0.0009***	0.0008***
	(3.021)	(2.257)	(2.412)	(3.008)	(3.525)	(3.525)	(4.798)
Control variables	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year, industry, and country FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Z	1917	1917	1917	1917	1917	1917	1917
R^2	0.1121	0.1038	0.1063	0.1110	0.1119	0.1119	0.1173
Panel B: Target gains							
Dependent variable:	(1)	(2)	(3)	(4)	(5)	(9)	(7)
	Kogut and Singh (1988)						
Target gains $(-1, +1)$	index	PDI	IVC	MVF	UAI	IVR	IVS
National cultural distance	0.0018***	0.0008***	0.0008***	0.0014***	0.0012***	0.0025***	0.0015***
	(4.044)	(3.006)	(3.029)	(4.290)	(4.988)	(4.988)	(4.430)
Control variables	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year, industry, and country FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	1917	1917	1917	1917	1917	1917	1917
R^2	0.1623	0.1506	0.1560	0.1645	0.1601	0.1601	0.1566
Letelmoon Tion 2 and a state of the state of	(1000 0000)	17-1-1	[4 . 4	2 - III			1 - 1 - 1

Note: The CBMA's sample of 1917 completed deals spans 19 years (2003-2021), encompassing both acquirers and targets that are public firms. The acquirer and target gains are calculated as cumulative abnormal returns (CARs) over a 3-day event window surrounding the deal announcement. CARs are calculated using the market model for the period of 255 to 25 days (estimation window) before the announcement day. The variable of interest (national cultural distance) is calculated using Kogut and Singh's (1988) index in Model (1), and from Models (2) to (7), we use six individual dimensions of Hofstede (2011). All models use the same control variables, which are reported in Table 3. Estatistics are shown in parentheses, and standard errors are corrected for heteroscedasticity (White 1980). We control for year, industry, and country fixed effects in each model and do not report their coefficients for brevity. 1468016, 0, Downloaded from https://onlinelibrary.wiley.com/doi/10.1111/fmii.70002 by INASPHINAR1 - CAMEROON, Wiley Online Library on [27/08/2025]. See the Terms and Conditions (https://onlinelibrary.wiley.com/erms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

^{*, **,} and *** show statistical significance levels at 10%, 5%, and 1%, respectively.

p < 0.1, *p < 0.05, **p < 0.01.

1468046, 0, Downloaded from https://onlinelibrary.wiley.com/doi/10.1111/fmii.70002 by INASPHINARI - CAMEROON, Wiley Online Library on [27/08/2025]. See the Terms and Conditions (https://onlinelibrary.wiley.com/rems-and-conditions) on Wiley Online Library for rules of use; OA arches are governed by the applicable Creative Commons Licensen and Conditions (https://onlinelibrary.wiley.com/rems-and-conditions) on Wiley Online Library for rules of use; OA arches are governed by the applicable Creative Commons Licensen and Conditions (https://onlinelibrary.wiley.com/rems-and-conditions) on Wiley Online Library for rules of use; OA arches are governed by the applicable Creative Commons Licensen and Conditions (https://onlinelibrary.wiley.com/rems-and-conditions) on Wiley Online Library for rules of use; OA arches are governed by the applicable Creative Commons Licensen and Conditions (https://onlinelibrary.wiley.com/rems-and-conditions) on Wiley Online Library for rules of use; OA arches are governed by the applicable Creative Commons Licensen and Conditions (https://onlinelibrary.wiley.com/rems-and-conditions) on Wiley Online Library for rules of use; OA arches are governed by the applicable Creative Commons Licensen and Conditions (https://onlinelibrary.wiley.com/rems-and-conditions) on Wiley Online Library for rules of use; OA arches are governed by the applicable Creative Commons and Conditions (https://onlinelibrary.wiley.com/rems-and-conditions) on Wiley Online Library for rules of use; OA arches are governed by the applicable Creative Commons and Conditions (https://onlinelibrary.wiley.com/rems-and-conditions) on Wiley Online Library for rules of use; OA arches are governed by the applicable Creative Commons are governed by the a

TABLE 5 | Boundary conditions—Stock payment and experienced acquirers.

	Moderating	ng effect		NS	Subsamples	
Dependent variable:	(1)	(2)	(3)	(4)	(5)	(9)
Synergistic gains $(-1, +1)$	Full sample	Full sample	Stock payment	Cash or mixed payment	Experienced acquirers	Non-experienced acquirers
National cultural distance (NCD)	0.0007***	0.0002	0.0025**	0.0002	0.0011***	0.0007
	(2.690)	(0.678)	(2.167)	(1.203)	(3.220)	(0.669)
Stock payment (SP)	0.0011					
	(0.131)					
$NCD \times SP$	0.0013*					
	(1.678)					
Experienced acquirer (EA)		-0.0078				
		(-0.935)				
$NCD \times EA$		0.0010*				
		(1.897)				
Control variables	Yes		Yes	Yes	Yes	Yes
Year, industry, and country FE	Yes		Yes	Yes	Yes	Yes
N	1917	1917	290	1627	253	1664
R^2	0.0935	0.0885	0.2764	0.1192	0.2905	0.1170

Note: This table presents regression results for the effect of national cultural distance on synergistic gains when the payment is made by issuing stock or when the acquirer is experienced (i.e., acquirers with more than three deals during the sample period). In Models (1) and (2), the variables of interest are interest are interaction terms (NCD × SP and NCD × EA). From Models (3) to (6), we report the effect of the national culture distance on synergistic gains for subsamples of deals of stock payments and experienced acquirers. We report t-statistics in parentheses and correct standard errors for heteroscedasticity (White 1980). All models control for year, industry, and country fixed effects, whose coefficients are not reported for brevity.

^{*, **} and *** show statistical significance at 10%, 5%, and 1%, respectively.

p < 0.1, p < 0.05, p < 0.05, p < 0.01.

TABLE 6 | Addressing endogeneity.

Panel A: PSM		
Dependent variable:	(1)	(2)
Synergistic gains (-1, +1)	Matched sample	Matched sample
National cultural distance	0.0260***	0.0010***
	(3.084)	(3.463)
Control variables	Yes	Yes
Year, industry, and country FE	Yes	Yes
N	1442	1917
R^2	0.1243	0.1126
Panel B: Heckman test		
Dependent variable:		(1)
Synergistic gains (-1, +1)		Kogut and Singh (1988) index
National cultural distance		0.0017***
		(4.619)
Heckman λ (Mill's ratio)		-0.1208***
		(-2.499)
Control variables		Yes
Year, industry, and country FE		Yes
N		3547
R^2		0.2594

Note: This table shows results for the impact of cultural distance on synergistic gains after addressing endogeneity-related issues using PSM (Model 1 of Panel A), EBM (Model 2 of Panel A), and the Heckman test (Panel B). The synergistic gain is the value-weighted portfolio of the acquirer and target gains over a 3-day event window. Our key variable of interest ('National cultural distance') is calculated using Kogut and Singh's (1988) index, which captures the difference between cultural scores of merging firms' home countries. All variables are defined in Appendix A; standard errors are corrected for heteroscedasticity (White 1980), and *t*-statistics are reported in parentheses. We control for year, industry, and country fixed effects in all models, whose coefficients are not reported for brevity.

*, ***, and *** show statistical significance levels at 10%, 5% and 1%, respectively.

TABLE 7 | Similarity in national culture and takeover gains.

	(1)	(2)	(3)
Dependent variables:	Synergistic gains	Acquirer gains	Target gains
National cultural similarity	-0.0819***	-0.0462***	-0.1501***
	(-3.023)	(-2.777)	(-3.537)
Control variables	Yes	Yes	Yes
Year, industry, and country FE	Yes	Yes	Yes
N	1917	1917	1917
R^2	0.1118	0.1083	0.1546

Note: The table reports regression results for the impact of national cultural similarity on synergistic gains, acquirer gains, and target gains using the sample of CBMAs. The combining firms are publicly listed, and we consider only completed deals. Our key variable of interest (national culture similarity) is calculated using Kogut and Singh's (1988) index. Synergistic gain is a value-weighted portfolio of the acquirer and target cumulative abnormal returns (CARs) over a 3-day event window. The CARs for combining firms are calculated using the market model for the period of 255 to 25 days before the deal announcement. Involved variables are defined in Appendix A; t-statistics are reported in parentheses, and standard errors are corrected for heteroscedasticity (White 1980). All models control for year, industry, and country fixed effects, whose coefficients are not reported for brevity.

p < 0.1, p < 0.05, p < 0.01.

^{*, **,} and *** show statistical significance levels at 10%, 5%, and 1%, respectively.

p < 0.1, p < 0.05, p < 0.01.

7 | Robustness Tests

In this section, we investigate the robustness of the impact of the national cultural distance on synergistic gains documented above and show all findings in Table 8.

First, we employed an alternative index of the national culture distance using nine cultural dimensions provided by GLOBE¹⁰ (Ahmad et al. 2022; House et al. 2004; Kashefi-Pour et al. 2020). These dimensions include gender egalitarianism, uncertainty avoidance, human orientation, performance orientation, assertiveness, power distance, institutional collectivism, in-group collectivism, and future orientation. As reported in Panel A of Table 8, our findings are in alignment with the prior analyses and show that synergistic gains are higher when the national cultural distance is higher (either using the index or individual dimensions).¹¹

Second, we used alternative windows for calculating returns (i.e., 5-, 7-, 9-, 11-, 21-, and 51-day) and developing value-weighted portfolios to make sure that the inferences drawn are not based on the shorter event window of 3 days. The results are qualitatively similar as reported in Panel B of Table 8.

Third, the year 2014, the business services industry, and the United States are dominant in our sample of CBMAs, and we present the evidence that our results are not driven by such dominance. After dropping deals from the dominant year, industry, and country, we still find a positive association between the national cultural distance and synergistic gains (see Panel C of Table 8).

Fourth, we follow existing studies¹² (Erel et al. 2012; Guiso et al. 2009) and control for additional control variables, including home and host countries' same language, level of trust, majority control deals, and deals where the acquirer is from a developed country and the target is from an emerging economy. The data on language and trust is from the World Value Survey, which is widely used in the literature (Ahern et al. 2015; Maung 2022). The majority control deal is a dummy variable that is equal to one if the acquirer has less than 50% stake in the target before the deal announcement and has more than 50% stake in the target¹³ after the successful deal, and zero otherwise. Also, the developed-toemerging economy deal is a dummy variable with the value of one if the acquirer is from the developed economy and the target is from an emerging economy, and zero otherwise. Our results are qualitatively similar to what we found previously, namely that cultural distance increases synergistic gains (see Model 1 of Panel D). Additionally, we use acquirer and target country-pair fixed effects and report results in Model (2) of Table 8.

Finally, following Madura and Ngo (2008), we use the takeover premium as an alternative proxy¹⁴ of takeover synergies and examine whether our baseline results still hold. The SDC database provides three proxies for the takeover premium and defines them as the ratio of the acquirer's offer price to the target's stock price 1 day (1-day premium), 1 week (1-week premium), and 4 weeks (4-week premium) before the deal announcement. Using all these proxies, we find that national cultural distance improves takeover synergies (Panel E of Table 8). Overall, our key findings

pass all robustness tests and suggest that the association between cultural distance and synergistic gains is causal.

8 | Discussion and Conclusions

8.1 | Summary

In this study, we fill an important research gap and test our hypotheses on how cultural distance can create higher synergistic gains. If CBMAs promote cultural diversity in the combined firm, we expect a positive association between the national cultural distance and synergistic gains. The results show that synergistic gains increase with the national cultural distance (H1), and this association becomes more pronounced for stock-financed deals and takeover experience (H2). Our results pass several robustness tests, including alternative measures of national culture and synergistic gains, subsamples, and controlling for selection bias. In all of our regression analyses, we control for the organizational cultural gap to make sure that our inferences are not caused by it.

The empirical evidence supports organizational learning theory—diverse cultural practices can be learnt—and the AOF concept, showing that acquirers can create takeover value by acquiring firms from countries with dissimilar cultural practices. The disparity in national cultural practices appears to be an important determinant of takeover synergies for CBMAs rather than a hindrance to synergies. We believe that organizational learning theory caters to new cognizance for the intersections of international business and finance research, as the relevant theory and empirical findings repudiate the established view that national cultural dissimilarities destroy takeover value.

8.2 | Implications for Managers and Policymakers

Our results provide insights for managers and policymakers on how the takeover market can serve as a vehicle to promote learning of diverse cultural practices between merging firms. As the disparity in national cultures creates takeover value, the managers of acquirer and target firms should make informed decisions regarding CBMAs. Combining firm managers needs a proper evaluation of cultural norms in the takeover process—from the deal initiation to finalization—for promoting cross-cultural understanding to minimize post-integration challenges, fostering effective management styles, promoting conflict resolution, and developing effective integration strategies. Our study can also help managers foster innovation in the combined firm because managers from different cultures may offer unique ideas.

Policymakers can use this study as a stepping stone for investigating how CBMAs can contribute to shareholder wealth. For firms involved in CBMAs, they must promote cultural training programmes, provide subsidies and tax incentives, encourage firms to establish diverse cultural boards and establish common standards for cultural harmonization. Considering these implications, policymakers can contribute to the success of CBMAs and foster international trade.

 TABLE
 8
 | Robustness tests.

Dependent variable:	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Synergistic gains (-1,										
+1)	Globe index	Globe 1	Globe 2	Globe 3	Globe 4	Globe 5	Globe 6	Globe 7	Globe 8	Globe 9
National cultural distance	0.0054***	0.0038**	0.0056***	0.0043***	0.0059***	0.0064***	0.0049***	0.0047***	0.0053**	0.0047***
	(3.389)	(2.271)	(3.023)	(3.005)	(3.581)	(3.283)	(2.913)	(2.637)	(2.385)	(2.605)
Control variables	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year, industry, and country FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
N	1732	1732	1732	1732	1732	1732	1732	1732	1732	1732
R^2	0.0412	0.0397	0.0410	0.0407	0.0414	0.0414	0.0407	0.0404	0.0399	0.0405
Panel B: Alte	rnative windo	ows of syne	rgistic gai	ns						
Dependent variable:		(1)		(2)		(3)		(4)		(5)
Synergistic gains	5-da	y gains	7-d	ay gains	11	-day gains	S	21-day gains		51-day gains
	(-	2, +2)	(-	-3, +3)		(-5, +5)		(-10, +10)	1	(-25, +25)
National cultural distance	0.0	012***	0.0	0010***	(0.0010***		0.0009***		0.0009***
Control variables		Yes		Yes		Yes				
Year, industry, and country FE		Yes		Yes		Yes		(3.510)		(3.496)
N		1917		1917		1917		1917		1917
R^2	0	.1466	(0.1576		0.1721		0.1743		0.1738
Panel C: Excl	uding domina	ınt year, in	dustry, and	d country						
Dependent va	ariable:		(1)			(2)			(3	3)
Synergistic ga	ains (-1, +1)	Excl	uding 2014	Exclu	ding Busi	ness Servi	ices indus	try Exc	cluding U	nited State
National cultu	ral distance	0.	0010***		(0.0013***			0.001	13***
			(3.415)			(3.451)			(4.2	231)
Control variab	les		Yes			Yes			Ye	es
Year, industry,	and country Fl	Е	Yes			Yes			Ye	es
N			1749			1592			114	45
R^2			0.1243			0.1207			0.10	691

(Continues)

TABLE 8 | (Continued)

Panel D: Additional controls and count	ry-pair fixed effects	
Dependent variable:	(1)	(2)
Synergistic gains $(-1, +1)$	Kogut and Singh (1988) index	Kogut and Singh (1988) index
National cultural distance	0.0010***	0.0120***
	(3.287)	(3.011)
Trust	0.0061	
	(0.869)	
Same language	0.0281	
	(1.158)	
Majority control	0.0189***	
	(2.594)	
Developed to emerging market	0.0091	
	(1.145)	
Year, industry, and country FE	Yes	Yes
Acquirer and target country-pair FE	No	Yes
N	1917	1917
R^2	0.1181	0.2106

Panel E: Alternative measures of t	akeover		
Dependent variable:	(1)	(2)	(3)
Takeover premium	1-day premium	1-week premium	4-week premium
National cultural distance	0.0007***	0.0005***	0.0006***
	(2.969)	(3.303)	(3.144)
Control variables	Yes	Yes	Yes
Year, industry, and country FE	Yes	Yes	Yes
N	1917	1917	1917
R^2	0.2273	0.2062	0.2234

Note: The table documents regression results for the impact of the national cultural distance on synergistic gains using alternative measures of national culture (Panel A), alternative windows of synergistic gains (Panel B), for different subsamples (Panel C), using additional control variables and country-pair fixed effects (Panel D), and using alternative measures of takeover synergies (Panel E). Our dependent variable is synergistic gains, and the key variable of interest is the national cultural distance. In Panel A, we utilized the GLOBE database to obtain scores (1 to 7) on nine cultural dimensions identified by House et al. (2004) as a proxy for national culture, rather than Hofstede's cultural dimensions. Panel B, in separate regressions, illustrates synergistic gains across 5-, 7-, 11-, 21-, and 51-day event windows surrounding the deal announcement. Panel C reports the results for the impact of national culture distance on synergistic gains using different subsamples, excluding deals from the top year, industry, and country. Panel D illustrates the impact of cultural distance on synergistic gains, after controlling for additional variables and using country-pair fixed effects. *t*-statistics are in parentheses, and robust standard errors are used to account for heteroscedasticity (White 1980). All models control for year, industry, and country fixed effects, whose coefficients are not reported for brevity. We define all variables in Appendix A.

*, **, **, and *** denote the statistical significance levels at 10%, 5%, and 1%, respectively.

8.3 | Limitations

This study is limited in fewer ways that open avenues for future research on cultural dissimilarities in CBMAs. Although our work covers an international sample of CBMAs, caution is needed in generalizing our findings to other restructuring activities such as joint ventures, strategic alliances, and spin-offs. Further work must explore the impacts of the national cultural distance on the outcomes of other restructuring activities.

Considering our empirical research design, we focus on CBMAs of publicly listed acquirers and targets with available data on all variables used in our study, and therefore our final sample is subject to debate when we want to expand its findings to the bigger CBMAs sample that includes private acquirers and targets. If the stock price data is available for private firms engaging in CBMAs, our established relationships can be tested using a bigger CBMAs sample.

p < 0.1, p < 0.05, p < 0.01.

We do not investigate how the association between the national cultural distance and synergistic gains is affected by the competitiveness of the takeover market. Future work should investigate how the takeover competition in the acquirer or target country affects the relationship between synergistic gains and cultural distance in CBMAs.

Conflicts of Interest

The authors declare no conflicts of interest.

Endnotes

- ¹Empirical affirmation endures indecisively, with findings ranging from no impact to U-shaped, negative, and positive impacts (Brock 2005; Rottig 2017; Stahl et al. 2013) of the cultural distance on takeover performance.
- ²Firms vary more in terms of national culture than organizational culture in mergers and acquisitions (see, among others, Bereskin et al. 2018). However, we control for organizational culture in all of our regression analyses to make sure that any inferences are not caused by organizational culture.
- ³To capture cultural distance, we used the cultural difference index proposed by Kogut and Singh (1988) based on Hofstede's (2011) six cultural dimensions.
- ⁴ Synergy motive in mergers and acquisitions (M&As) suggests that the value of the combined firm is higher than the individual values of the combining firms (Berkovitch and Narayanan 1993). Following Bradley et al. (1988), we calculate synergistic gains by developing the value-weighted portfolio of combining firms where weights are allocated based on the equity's market value 6 days prior to the announcement day. Also, similar to Wang and Xie (2009), we adjust target weighted returns for the toehold.
- ⁵In robustness tests, we use alternative measures of national cultural distance and synergistic gains and find qualitatively similar results.
- ⁶ Although the business services industry dominates our sample, the results for national cultural distance are persistent after excluding deals from this industry (see Panel C of Table 8).
- ⁷In robustness tests, we used 5-, 7-, 11-, 21-, and 51-day event windows to calculate the value-weighted portfolio of the acquirer and target.
- ⁸To review the determinants of M&As and their wealth effects, see Martynova and Renneboog (2008) and Jensen and Ruback (1983).
- ⁹In Model (1) of Table 3, the coefficient on national cultural distance is 0.0010 (*t*-statistic of 3.172) with 17.56 standard deviation. So, an increase of one standard deviation in national cultural distance increases synergistic gains by 1.75 percentage points (standard deviation \times β coefficient \times 100 = 17.56 \times 0.0010 \times 100 = 1.75).
- ¹⁰GLOBE culture dimensions are based on surveys from 62 nations around the world.
- ¹¹We dropped 2 acquirer countries and 20 target countries because GLOBE does not cover these countries. They are Belarus, Bulgaria, Croatia, Estonia, Ethiopia, Iceland, Latvia, Lithuania, Luxembourg, Malta, Moldova, Norway, Belgium, Panama, Peru, Puerto Rico, Romania, Saudi Arabia, Serbia, the Slovak Republic, Ukraine, and Vietnam.
- $^{12}\,\mathrm{We}$ are thankful to the anonymous reviewer for drawing our attention to other determinants of takeover performance.
- ¹³This is a standard definition of majority control deals in the finance literature (see, among others, Hussain and Shams 2022).
- ¹⁴We are grateful to the anonymous reviewer for highlighting alternative measures of takeover synergies.

References

Ahmad, M. F., S. Aziz, and M. M. Dowling. 2022. "Does Target Country Cultural Orientation Influence M&A?" *British Journal of Management* 33, no. 2: 906–938.

Ahammad, M. F., V. Leone, S. Y. Tarba, K. W. Glaister, and A. Arslan. 2017. "Equity Ownership in Cross-Border Mergers and Acquisitions by British Firms: An Analysis of Real Options and Transaction Cost Factors." *British Journal of Management* 28, no. 2: 180–196.

Ahern, K. R., D. Daminelli, and C. Fracassi. 2015. "Lost in Translation? The Effect of Cultural Values on Mergers Around the World." *Journal of Financial Economics* 117, no. 1: 165–189.

Ahammad, M. F., and K. W. Glaister. 2013. "The Pre-Acquisition Evaluation of Target Firms and Cross Border Acquisition Performance." *International Business Review* 22, no. 5: 894–904.

Aktas, N., E. De Bodt, and R. Roll. 2011. "Serial Acquirer Bidding: An Empirical Test of the Learning Hypothesis." *Journal of Corporate Finance* 17, no. 1: 18–32.

Andrade, G., M. Mitchell, and E. Stafford. 2001. "New evidence and perspectives on mergers." *Journal of economic perspectives* 15, no. 2: 103–120.

Arouri, M., M. Gomes, and K. Pukthuanthong. 2019. "Corporate social responsibility and M&A uncertainty." *Journal of Corporate Finance* 56: 176–198.

Alexandridis, G., C. F. Mavrovitis, and N. G. Travlos. 2012. "How Have M&As Changed? Evidence From the Sixth Merger Wave." *European Journal of Finance* 18, no. 8: 663–688.

Alexandridis, G., D. Petmezas, and N. G. Travlos. 2010. "Gains From Mergers and Acquisitions Around the World: New Evidence." *Financial Management* 39, no. 4: 1671–1695.

Aybar, B., and A. Ficici. 2009. "Cross-Border Acquisitions and Firm Value: An Analysis of Emerging-Market Multinationals." *Journal of International Business Studies* 40: 1317–1338.

Bereskin, F., S. K. Byun, M. S. Officer, and J. M. Oh. 2018. "The Effect of Cultural Similarity on Mergers and Acquisitions: Evidence From Corporate Social Responsibility." *Journal of Financial and Quantitative Analysis* 53, no. 5: 1995–2039.

Berkovitch, E., and M. P. Narayanan. 1993. "Motives for Takeovers: An Empirical Investigation." *Journal of Financial and Quantitative Analysis* 28, no. 3: 347–362.

Billett, M. T., and Y. Qian. 2008. "Are Overconfident CEOs Born or Made? Evidence of Self-Attribution Bias From Frequent Acquirers." *Management Science* 54, no. 6: 1037–1051.

Björkman, I., G. K. Stahl, and E. Vaara. 2007. "Cultural Differences and Capability Transfer in Cross-Border Acquisitions: The Mediating Roles of Capability Complementarity, Absorptive Capacity, and Social Integration." *Journal of International Business Studies* 38, no. 4: 658–672.

Boeh, K. K. 2011. "Contracting Costs and Information Asymmetry Reduction in Cross-Border M&A." *Journal of Management Studies* 48, no. 3: 568–590.

Boone, A. L., E. Lie, and Y. Liu. 2014. "Time Trends and Determinants of the Method of Payment in M&As." *Journal of Corporate Finance* 27: 296–304.

Boone, A. L., and J. H. Mulherin. 2007. "Do termination provisions truncate the takeover bidding process?." *The Review of Financial Studies* 20, no. 2: 461–489.

Bradley, M., A. Desai, and E. H. Kim. 1988. "Synergistic Gains From Corporate Acquisitions and Their Division Between the Stockholders of Target and Acquiring Firms." *Journal of Financial Economics* 21, no. 1: 3–40.

Brigida, M., J. Madura, and A. Viale. 2014. "An Information-Based Model of Target Stock Price Runup in the Market for Corporate Control." *Quantitative Finance* 14, no. 6: 1019–1030.

- Brock, D. M. 2005. "Multinational Acquisition Integration: The Role of National Culture in Creating Synergies." *International Business Review* 14, no. 3: 269–288.
- Campagnolo, D., and G. Vincenti. 2022. "Cross-Border M&As: The Impact of Cultural Friction and CEO Change on the Performance of Acquired Companies." *Journal of International Management* 28, no. 4: 100942.
- Chand, A., R. Jiang, and Z. Zhao. 2021. "National Cultural Distance and Cross-Border M&A." *Asia-Pacific Journal of Accounting & Economics* 30, no. 5: 1350–1367.
- Chao, Y. C. 2018. "Organizational Learning and Acquirer Performance: How Do Serial Acquirers Learn From Acquisition Experience?" *Asia Pacific Management Review* 23, no. 3: 161–168.
- Chari, M. D., and K. Chang. 2009. "Determinants of the Share of Equity Sought in Cross-Border Acquisitions." *Journal of International Business Studies* 40: 1277–1297.
- Chatterjee, S., I. Hasan, K. John, and A. Yan. 2021. "Stock Liquidity, Empire Building, and Valuation." *Journal of Corporate Finance* 70: 102051.
- Chikhouni, A., G. Edwards, and M. Farashahi. 2017. "Psychic Distance and Ownership in Acquisitions: Direction Matters." *Journal of International Management* 23, no. 1: 32–42.
- Col, B. 2017. "Agency Costs of Moving to Tax Havens: Evidence From Cross-Border Merger Premia." *Corporate Governance: An International Review* 25, no. 4: 271–288.
- Cornett, M. M., B. Tanyeri, and H. Tehranian. 2011. "The Effect of Merger Anticipation on Bidder and Target Firm Announcement Period Returns." *Journal of Corporate Finance* 17, no. 3: 595–611.
- Cuervo-Cazurra, A., M. M. Maloney, and S. Manrakhan. 2007. "Causes of the Difficulties in Internationalization." *Journal of International Business Studies* 38: 709–725.
- Curado, C. 2006. "Organisational Learning and Organisational Design." *Learning Organization* 13, no. 1: 25–48.
- Cuypers, I. R., Y. Cuypers, and X. Martin. 2017. "When the Target May Know Better: Effects of Experience and Information Asymmetries on Value From Mergers and Acquisitions." *Strategic Management Journal* 38, no. 3: 609–625.
- Danbolt, J., A. Siganos, and A. Tunyi. 2016. "Abnormal Returns From Takeover Prediction Modelling: Challenges and Suggested Investment Strategies." *Journal of Business Finance & Accounting* 43, no. 1–2: 66–97.
- Dang, M., V. A. Hoang, E. Jones, D. Henry, P. U. Le, and P. Puwanenthiren. 2022. "Country Uncertainty, Power Distance, and Payment Methods in Acquisitions." *European Journal of Finance* 28, no. 16: 1541–1570.
- Dhanaraj, C., and P. W. Beamish. 2004. "Effect of Equity Ownership on the Survival of International Joint Ventures." *Strategic Management Journal* 25, no. 3: 295–305.
- Doukas, J. A., and R. Zhang. 2021. "Managerial Ability, Corporate Social Culture, and M&As." *Journal of Corporate Finance* 68: 101942.
- Drejer, A. 2000. "Organisational Learning and Competence Development." *Learning Organization* 7, no. 4: 206–220.
- Ellis, J. A., S. B. Moeller, F. P. Schlingemann, and R. M. Stulz. 2017. "Portable Country Governance and Cross-Border Acquisitions." *Journal of International Business Studies* 48: 148–173.
- Erel, I., R. C. Liao, and M. S. Weisbach. 2012. "Determinants of Cross-Border Mergers and Acquisitions." *Journal of Finance* 67, no. 3: 1045–1082.
- Fama, E. F., L. Fisher, M. C. Jensen, and R. Roll. 1969. "The Adjustment of Stock Prices to New Information." *International Economic Review* 10, no. 1: 1–21.
- Fauver, L., G. Loureiro, and A. G. Taboada. 2017. "The Impact of Regulation on Information Quality and Performance Around Seasoned Equity Offerings: International Evidence." *Journal of Corporate Finance* 44: 73–98.

- Fuller, K., J. Netter, and M. Stegemoller. 2002. "What Do Returns to Acquiring Firms Tell Us? Evidence From Firms That Make Many Acquisitions." *Journal of Finance* 57, no. 4: 1763–1793.
- Gaughan, P. A. 2002. "What's the Outlook for M&A in an Uncertain Market?." *The Journal of Corporate Accounting & Finance* 13, no. 5: 51.
- Gaur, A. S., and J. W. Lu. 2007. "Ownership Strategies and Survival of Foreign Subsidiaries: Impacts of Institutional Distance and Experience." *Journal of Management* 33, no. 1: 84–110.
- Gilson, S. C. 1990. "Bankruptcy, Boards, Banks, and Blockholders: Evidence on Changes in Corporate Ownership and Control When Firms Default." *Journal of Financial Economics* 27, no. 2: 355–387.
- Golubov, A., D. Petmezas, and N. G. Travlos. 2012. "When It Pays to Pay Your Investment Banker: New Evidence on the Role of Financial Advisors in M&As." *Journal of Finance* 67, no. 1: 271–311.
- Goold, M., and A. Campbell. 1998. "Desperately Seeking Synergy." *Harvard Business Review* 76, no. 5: 131–143.
- Graebner, M. E., K. H. Heimeriks, Q. N. Huy, and E. Vaara. 2017. "The Process of Postmerger Integration: A Review and Agenda for Future Research." *Academy of Management Annals* 11, no. 1: 1–32.
- Graham, J. R., M. L. Lemmon, and J. G. Wolf. 2002. "Does Corporate Diversification Destroy Value?" *Journal of Finance* 57, no. 2: 695–720.
- Guiso, L., P. Sapienza, and L. Zingales. 2009. "Cultural Biases in Economic Exchange?" *Quarterly Journal of Economics* 124, no. 3: 1095–1131.
- Gupta, A. K., and V. Govindarajan. 2000. "Knowledge Flows Within Multinational Corporations." *Strategic Management Journal* 21, no. 4: 473–496.
- Hainmueller, J., and D. Hangartner. 2013. "Who gets a Swiss passport? A natural experiment in immigrant discrimination." *American political science review* 107, no. 1: 159–187.
- Hansen, R. G. 1987. "A Theory for the Choice of Exchange Medium in Mergers and Acquisitions." *Journal of Business* 60, no. 1: 75–95.
- Heckman, J. J. 1979. "Sample Selection Bias as a Specification Error." *Econometrica: Journal of the Econometric Society* 47, no. 1: 153–161.
- Hitt, M. A., M. T. Dacin, B. B. Tyler, and D. Park. 1997. "Understanding the Differences in Korean and US Executives' strategic Orientations." *Strategic Management Journal* 18, no. 2: 159–167.
- Hoffman, E., K. McCabe, and V. L. Smith. 1996. "Social Distance and Other-Regarding Behavior in Dictator Games." *American Economic Review* 86, no. 3: 653–660.
- Hofstede, G. 1980. "Motivation, Leadership, and Organization: Do American Theories Apply Abroad?" *Organizational Dynamics* 9, no. 1: 42–63.
- Hofstede, G. 2001. Culture's Consequences: Comparing Values, Behaviors, Institutions and Organizations Across Nations. SAGE Publications.
- Hofstede, G. 2011. "Dimensionalizing Cultures: The Hofstede Model in Context." *Online Readings in Psychology and Culture* 2, no. 1: 1–26.
- House, R. J., P. J. Hanges, M. Javidan, P. W. Dorfman, and V. Gupta, eds. 2004. *Culture, Leadership, and Organizations: The GLOBE Study of 62 Societies*. SAGE Publications.
- Huang, P., M. S. Officer, and R. Powell. 2016. "Method of Payment and Risk Mitigation in Cross-Border Mergers and Acquisitions." *Journal of Corporate Finance* 40: 216–234.
- Hussain, T., and N. Kumar. 2025. "How Do Green Acquirers Select Targets? Value of Green Innovation in Takeovers." *British Journal of Management* 36, no. 3: 1303–1325.
- Hussain, T., and G. Loureiro. 2023. "Target Industry Takeover Competition and the Wealth Effects of Mergers and Acquisitions: International Evidence." *Journal of International Financial Markets, Institutions and Money* 89: 101865.
- Hussain, T., and S. Shams. 2022. "Pre-Deal Differences in Corporate Social Responsibility and Acquisition Performance." *International Review of Financial Analysis* 81: 102083.

Hussain, T., and A. A. Tunyi. 2025. "Do Environmental, Social, and Governance Standards Improve the Bargaining Power of Bidders? An Empirical Investigation." *Journal of Environmental Management* 373: 123468.

Hussain, T., A. A. Tunyi, M. Sufyan, and Y. Shahab. 2022. "Powerful Bidders and Value Creation in M&As." *International Review of Financial Analysis* 81: 102076.

Hussain, T., A. A. Tunyi, and G. Areneke. 2024. "Environmental Innovation and Takeover Performance." *Business Strategy and the Environment* 33, no. 7: 6586–6615.

Jaffe, A. B. 1986. "Technological Opportunity and Spillovers of R&D: Evidence From Firms' Patents, Profits and Market Value." Working Paper No. 1815. NBER.

Jaffe, J., D. Pedersen, and T. Voetmann. 2013. "Skill Differences in Corporate Acquisitions." *Journal of Corporate Finance* 23: 166–181.

Jemison, D. B., and S. B. Sitkin. 1986. "Corporate Acquisitions: A Process Perspective." *Academy of Management Review* 11, no. 1: 145–163.

Jensen, M. C., and R. S. Ruback. 1983. "The market for corporate control: The scientific evidence." *Journal of Financial economics* 11, no. 1-4: 5–50.

Jiang, F., K. John, C. W. Li, and Y. Qian. 2018. "Earthly Reward to the Religious: Religiosity and the Costs of Public and Private Debt." *Journal of Financial and Quantitative Analysis* 53, no. 5: 2131–2160.

Kanungo, R. P. 2021. "Payment Choice of M&As: Financial Crisis and Social Innovation." *Industrial Marketing Management* 97: 97–114.

Kashefi-Pour, E., S. Amini, M. Uddin, and D. Duxbury. 2020. "Does Cultural Difference Affect Investment–Cash Flow Sensitivity? Evidence From OECD Countries." *British Journal of Management* 31, no. 3: 636–658.

Kogut, B., and H. Singh. 1988. "The Effect of National Culture on the Choice of Entry Mode." *Journal of International Business Studies* 19: 411–432.

Konara, P., and A. Mohr. 2023. "Cultural Bridging and the Performance of International Joint Ventures." *International Business Review* 32, no. 4: 102109.

Kroon, D. P., J. P. Cornelissen, and E. Vaara. 2015. "Explaining Employees' Reactions Towards a Cross-Border Merger: The Role of English Language Fluency." *Management International Review* 55: 775–800.

Krug, J. A., and D. Nigh. 2001. "Executive Perceptions in Foreign and Domestic Acquisitions: An Analysis of Foreign Ownership and Its Effect on Executive Fate." *Journal of World Business* 36, no. 1: 85–105.

Laamanen, T., and T. Keil. 2008. "Performance of Serial Acquirers: Toward an Acquisition Program Perspective." *Strategic Management Journal* 29, no. 6: 663–672.

Lang, L. H., R. Stulz, and R. A. Walkling. 1991. "A Test of the Free Cash Flow Hypothesis: The Case of Bidder Returns." *Journal of Financial Economics* 29, no. 2: 315–335.

Lee, J. M., Y. Paik, S. Horak, and I. Yang. 2022. "Turning a Liability Into an Asset of Foreignness: Managing Informal Networks in Korea." *Business Horizons* 65, no. 3: 351–364.

López-Duarte, C., and M. M. Vidal-Suárez. 2010. "External Uncertainty and Entry Mode Choice: Cultural Distance, Political Risk and Language Diversity." *International Business Review* 19, no. 6: 575–588.

Lubatkin, M., R. Calori, P. Very, and J. F. Veiga. 1998. "Managing Mergers Across Borders: A Two-Nation Exploration of a Nationally Bound Administrative Heritage." *Organization Science* 9, no. 6: 670–684.

Luypaert, M., and T. Van Caneghem. 2017. "Exploring the Double-Sided Effect of Information Asymmetry and Uncertainty in Mergers and Acquisitions." *Financial Management* 46, no. 4: 873–917.

Madura, J., and T. Ngo. 2008. "Clustered Synergies in the Takeover Market." *Journal of Financial Research* 31, no. 4: 333–356.

Mallon, M. R., and S. Fainshmidt. 2017. "Assets of Foreignness: A Theoretical Integration and Agenda for Future Research." *Journal of International Management* 23, no. 1: 43–55.

Martynova, M., and L. Renneboog. 2008. "Spillover of Corporate Governance Standards in Cross-Border Mergers and Acquisitions." *Journal of Corporate Finance* 14, no. 3: 200–223.

Masulis, R. W., C. Wang, and F. Xie. 2007. "Corporate Governance and Acquirer Returns." *Journal of Finance* 62, no. 4: 1851–1889.

Maung, M. 2022. "Trust and Cross-Border Mergers and Acquisitions." International Review of Financial Analysis 83: 102262.

Minkov, M., and G. Hofstede. 2011. "The Evolution of Hofstede's Doctrine." *Cross Cultural Management: An International Journal* 18, no. 1: 10–20.

Moeller, S. B., and F. P. Schlingemann. 2005. "Global Diversification and Bidder Gains: A Comparison Between Cross-Border and Domestic Acquisitions." *Journal of Banking & Finance* 29, no. 3: 533–564.

Morck, R., A. Shleifer, and R. W. Vishny. 1990. "Do Managerial Objectives Drive Bad Acquisitions?" *Journal of Finance* 45, no. 1: 31–48.

Morosini, P., S. Shane, and H. Singh. 1998. "National Cultural Distance and Cross-Border Acquisition Performance." *Journal of International Business Studies* 29: 137–158.

Mukherji, A., J. Mukherji, C. Dibrell, and J. D. Francis. 2013. "Overbidding in Cross-Border Acquisitions: Misperceptions in Assessing and Valuing Knowledge." *Journal of World Business* 48, no. 1: 39–46.

Myers, S. C., and N. S. Majluf. 1984. "Corporate Financing and Investment Decisions When Firms Have Information That Investors Do Not Have." *Journal of Financial Economics* 13, no. 2: 187–221.

Park, S. H., and G. R. Ungson. 1997. "The Effect of National Culture, Organizational Complementarity, and Economic Motivation on Joint Venture Dissolution." *Academy of Management Journal* 40, no. 2: 279–307.

Petersen, B., T. Pedersen, and M. A. Lyles. 2008. "Closing Knowledge Gaps in Foreign Markets." *Journal of International Business Studies* 39: 1097–1113.

Reddy, R. K., and F. Fabian. 2020. "Information Asymmetry and Host Country Institutions in Cross-Border Acquisitions." *Management International Review* 60, no. 6: 909–938.

Renneboog, L., and C. Vansteenkiste. 2019. "Failure and Success in Mergers and Acquisitions." *Journal of Corporate Finance* 58: 650–699.

Reus, T. H., and B. T. Lamont. 2009. "The Double-Edged Sword of Cultural Distance in International Acquisitions." *Journal of International Business Studies* 40: 1298–1316.

Rhodes-Kropf, M., and S. Viswanathan. 2004. "Market Valuation and Merger Waves." *Journal of Finance* 59, no. 6: 2685–2718.

 $Rossi, S., and P. F. Volpin. 2004. ``Cross-Country Determinants of Mergers and Acquisitions." {\it Journal of Financial Economics 74}, no. 2: 277–304.$

Rottig, D. 2017. "Meta-Analyses of Culture's Consequences for Acquisition Performance: An Examination of Statistical Artifacts, Methodological Moderators and the Context of Emerging Markets." *International Journal of Emerging Markets* 12, no. 1: 8–37.

Schwert, G. W. 1996. "Markup Pricing in Mergers and Acquisitions." *Journal of Financial Economics* 41, no. 2: 153–192.

Sethi, D., and W. Judge. 2009. "Reappraising Liabilities of Foreignness Within an Integrated Perspective of the Costs and Benefits of Doing Business Abroad." *International Business Review* 18, no. 4: 404–416.

Shahab, Y., C. X. Wang, P. E. Yeung, and J. N. Zhou. 2024. "The National Team: Stock Market Interventions and Corporate Catering Behavior." *International Review of Financial Analysis* 93: 103203.

Shi, T., J. Li, and C. L. Lim. 2017. "Host Country Consumers' Brand Attitudes After Cross-Border Acquisitions." *Journal of Product & Brand Management* 26, no. 6: 559–572.

- Shleifer, A., and R. W. Vishny. 2003. "Stock Market Driven Acquisitions." *Journal of Financial Economics* 70, no. 3: 295–311.
- Slangen, A. H. 2006. "National Cultural Distance and Initial Foreign Acquisition Performance: The Moderating Effect of Integration." *Journal of World Business* 41, no. 2: 161–170.
- Slangen, A. H., and J. F. Hennart. 2008. "Do Multinationals Really Prefer to Enter Culturally Distant Countries Through Greenfields Rather Than Through Acquisitions? The Role of Parent Experience and Subsidiary Autonomy." *Journal of International Business Studies* 39: 472–490.
- Stahl, G. K., and R. L. Tung. 2015. "Towards a More Balanced Treatment of Culture in International Business Studies: The Need for Positive Crosscultural Scholarship." *Journal of International Business Studies* 46: 391–414.
- Stahl, G. K., R. L. Tung, T. Kostova, and M. Zellmer-Bruhn. 2016. "Widening the Lens: Rethinking Distance, Diversity, and Foreignness in International Business Research Through Positive Organizational Scholarship." *Journal of International Business Studies* 47: 621–630.
- Stahl, G. K., and A. Voigt. 2008. "Do Cultural Differences Matter in Mergers and Acquisitions? A Tentative Model and Examination." *Organization Science* 19, no. 1: 160–176.
- Stahl, G. K., D. N. Angwin, P. Very, et al. 2013. "Sociocultural integration in mergers and acquisitions: Unresolved paradoxes and directions for future research." *Thunderbird international business review* 55, no. 4: 333–356.
- Starks, L. T., and K. D. Wei. 2013. "Cross-Border Mergers and Differences in Corporate Governance." *International Review of Finance* 13, no. 3: 265–297.
- Tajfel, H., and J. C. Turner. 2004. "The Social Identity Theory of Intergroup Behavior." In *Political Psychology*, edited by J. T. Jost and J. Sidanius, 276–293. Psychology Press.
- Tan, D., and J. T. Mahoney. 2003. "Explaining the Utilization of Managerial Expatriates From the Perspectives of Resource-Based, Agency, and Transaction-Costs Theories." In Managing Multinationals in a Knowledge Economy: Economics, Culture, and Human Resources: 15 (Advances in International Management), edited by J. L. C. Cheng and M. A. Hitt, 179–205. Emerald Group Publishing Limited.
- Taussig, M. 2017. "Foreignness as Both a Global Asset and a Local Liability: How Host Country Idiosyncrasies and Business Activities Matter." *Journal of International Business Studies* 48: 498–522.
- Tunyi, A. A., T. Hussain, G. Areneke, and J. Agyemang. 2025. "Co-Opted Boards and the Obfuscation of Financial Reports." Abacus, ahead of print, March 20.
- Very, P., M. Lubatkin, and R. Calori. 1996. "A Cross-National Assessment of Acculturative Stress in Recent European Mergers." *International Studies of Management & Organization* 26, no. 1: 59–86.
- Wang, C., and F. Xie. 2009. "Corporate Governance Transfer and Synergistic Gains From Mergers and Acquisitions." *Review of Financial Studies* 22, no. 2: 829–858.
- Wang, D., D. S. Hain, J. Larimo, and L. T. Dao. 2020. "Cultural Differences and Synergy Realization in Cross-Border Acquisitions: The Moderating Effect of Acquisition Process." *International Business Review* 29, no. 3: 101675.
- Weber, Y., and S. Y. Tarba. 2012. "Mergers and Acquisitions Process: The Use of Corporate Culture Analysis." *Cross Cultural Management: An International Journal* 19, no. 3: 288–303.
- White, H. 1980. "A heteroskedasticity-consistent covariance matrix estimator and a direct test for heteroskedasticity." *Econometrica: journal of the Econometric Society*, 817–838.
- Wu, Z., and R. Salomon. 2017. "Deconstructing the Liability of Foreignness: Regulatory Enforcement Actions Against Foreign Banks." *Journal of International Business Studies* 48: 837–861.

Appendix A Variable Definitions

Synergistic, acquirer, and target gains:

- (1) Synergistic gains
- (2) Acquirer and target gains

Pre-deal differences:

- (3) National cultural distance
- (4) Firm culture gap
- (5) National governance gap

Deal characteristics:

- (6) Related industry
- (7) Payment method
- (8) Number of competing acquirers

Acquirer and target characteristics:

- (9) Run-up
- (10) Leverage
- (11) Liquidity
- (12) Tobin Q

Acquirer country characteristics: (13)

GDP growth

(14) Log GDP per capita

- (1) The weighted average portfolio of the acquirer and target gains over 3-, 5-, 7-, 11-, 21-, and 51-day event windows around the day of the deal announcement. The weights are allocated based on the equity's market value 6 days before the deal announcement. Considering the toehold effect, we also adjust target weighted gains. Sources: Securities Data Corporation (SDC) and DataStream.
- (2) 3-, 5-, 7-, 11-, 21-, and 5-day cumulative abnormal returns (CARs) around the day of announcement using the market model for the estimation window of 255 to 25 days before the deal. Source: DataStream.
- (3) The difference in country-level culture (proxied by Hofstede's cultural dimensions) between the acquirer and target home countries. Source:

https://geerthofstede.com/research-and-vsm/dimension-data-matrix/

- (4) The difference in the firm-level culture (proxied by corporate social responsibility) scores between the acquirer and target firms. Source: ASSET4 ESG.
- (5) The difference in the country-level governance indices between the home countries of acquirers and targets. These indices are based on the average of six World Governance Indicators. Source: World Bank.
- (6) Dummy variable: 1 for related industry deal and 0 otherwise. Source: SDC.
- (7) Dummy variable: 1 for the purely stock-financed deal and 0 otherwise. Source: SDC.
- (8) Natural logarithm of the total number of acquirers involved in a cross-border deal. Source: SDC.
- (9) For synergistic gains regressions, we used a weighted average of the acquirer and target stock price run-up, where weights are allocated based on equity's market value. For acquirer and target gains regressions, we calculate their stock price run-up as the sum of abnormal returns over the 90 to 20 days preceding the deal announcement. Source: DataStream.
- (10) Long-term debt scaled by total assets. Source: WorldScope.
- (11) Cash and short-term investments scaled by total assets. Source: WorldScope.
- (12) (Total assets minus book value of equity plus market value of equity) divided by total assets. Source: WorldScope.
- (13) Annual growth in the real gross domestic product (GDP). Source: World Development Indicators.
- (14) Logarithm of real GDP (current US dollars divided by average population). Source: World Development Indicators.

Appendix B
Sample Distribution by the Target Country

Target country	N	%
Argentina	11	0.57
Australia	80	4.17
Austria	12	0.63
Bangladesh	3	0.16
Belarus	2	0.10
Belgium	33	1.72
Brazil	91	4.75
Bulgaria	8	0.42
Canada	149	7.77
Colombia	12	0.63

(Continues)

are governed by the applicable Creative Commons License

Target country	N	%
Costa Rica	1	0.05
Croatia	1	0.05
Denmark	24	1.25
Ecuador	2	0.10
Egypt	5	0.26
El Salvador	1	0.05
Estonia	1	0.05
Ethiopia	1	0.05
Finland	19	0.99
France	98	5.11
Georgia	2	0.10
Germany	114	5.95
Hong Kong	6	0.31
Hungary	4	0.21
Iceland	1	0.05
India	63	3.29
Indonesia	2	0.10
Israel	43	2.24
Italy	58	3.03
Japan	14	0.73
Latvia	2	0.10
Lithuania	5	0.26
Luxembourg	4	0.21
Malaysia	6	0.31
Malta	1	0.05
Mexico	9	0.47
Moldova	1	0.05
Morocco	7	0.37
Netherlands	72	3.76
New Zealand	12	0.63
Nigeria	3	0.16
Norway	36	1.88
Panama	1	0.05
Peru	7	0.37
Poland	19	0.99
Portugal	7	0.37
Puerto Rico	2	0.10
Romania	5	0.26
Saudi Arabia	4	0.21
Serbia	2	0.10
Singapore	15	0.78
Slovak Rep	3	0.16
		(Continues)

(Continues)

Target country	N	%
Slovenia	1	0.05
South Africa	22	1.15
Spain	48	2.50
Sweden	57	2.97
Switzerland	57	2.97
Thailand	3	0.16
Ukraine	1	0.05
United Kingdom	220	11.48
United States	418	21.80
Venezuela	1	0.05
Vietnam	5	0.26
Total	1917	100.00

Note: The table reports the sample distribution by the target firms' country, including the total number of deals (N) and the percentage (%) of deals over the 19-year sample period.

Appendix C Correlation Matrix

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)
(1)	1.00															
(2)	0.11*	1.00														
(3)	-0.01	0.07*	1.00													
(4)	0.04	0.10*	0.44*	1.00												
(5)	-0.04	0.07*	0.05	0.05	1.00											
(6)	0.06*	0.07*	-0.11*	-0.03	0.01	1.00										
(7)	-0.01	-0.02	-0.05	-0.02	-0.04	0.13*	1.00									
(8)	-0.04	0.03	0.03	0.05	0.05	0.00	0.05	1.00								
(9)	0.00	-0.06	-0.02	-0.09*	0.03	-0.01	0.01	0.02	1.00							
(10)	0.06*	-0.04	-0.27 *	-0.21*	0.03	0.04	0.14*	0.10*	0.11*	1.00						
(11)	-0.04	-0.01	0.02	0.01	0.04	-0.02	-0.02	0.00	-0.02	-0.06	1.00					
(12)	-0.04	0.15*	0.08*	-0.03	0.00	-0.01	-0.01	-0.05	-0.02	-0.08*	0.09*	1.00				
(13)	0.03	0.02	0.00	0.11*	0.06	-0.03	0.03	0.06	-0.03	-0.03	0.28*	-0.13 *	1.00			
(14)	-0.05	-0.01	-0.05	-0.05	-0.02	0.02	-0.01	-0.03	0.04	0.02	-0.04	0.04	-0.14 *	1.00		
(15)	0.03	-0.01	-0.06 *	-0.03	-0.06 *	-0.03	0.01	-0.01	-0.04	0.00	0.06*	-0.01	-0.06	-0.04	1.00	
(16)	-0.02	-0.09*	0.08*	0.03	-0.01	-0.03	-0.03	-0.04	0.09*	-0.11 *	0.32*	0.32*	0.23*	-0.03	-0.07 *	1.00

Note: The sample comprises 1,917 completed CBMAs reported in Securities Data Corporation (SDC) over 19 years (2003–2021), where the combining firms are publicly listed. The table reports correlation (along with level of significance) among all variables used in our regression analyses. (1) Synergistic gain. (2) National cultural distance. (3) Firm culture gap. (4) National governance gap. (5) Related industry dummy. (6) Payment method dummy. (7) Number of competing acquirers. (8) Run-up. (9) Acquirer leverage. (10) Target leverage. (11) Acquirer liquidity. (12) Target liquidity. (13) Acquirer Tobin Q. (14) Target Tobin Q. (15) GDP growth. (16) GDP per capita.

^{*, **,} and *** show statistical significance at 10%, 5%, and 1%, respectively. All variables are defined in Appendix A.